RESEARCH ARTICLE

Long-term cardiology outcomes in children after early treatment for Chagas disease, an observational study

Nicolás Leonel González¹*, Guillermo Moscatelli^{1,2}, Samanta Moroni¹, Griselda Ballering¹, Laura Jurado^{1,2}, Nicolás Falk¹, Andrés Bochoeyer³, Alejandro Goldsman³, María Grippo³, Héctor Freilij¹, Facundo Garcia Bournissen⁴, Eric Chatelain⁵, Jaime Altcheh^{1,2}

Servicio de Parasitología y Chagas, Hospital de Niños Ricardo Gutiérrez, Buenos Aires, Argentina,
Instituto Multidisciplinario de Investigación en Patologías Pediátricas (IMIPP) (CONICET-GCBA), Buenos Aires, Argentina, 3 Servicio de Cardiología, Hospital de Niños Ricardo Gutiérrez, Buenos Aires, Argentina,
Division of Pediatric Clinical Pharmacology, Department of Pediatrics, Schulich School of Medicine and Dentistry, University of Western Ontario, London, Ontario, Canada, 5 Drugs for Neglected Diseases initiative (DNDi), Geneva, Switzerland

* ng211@hotmail.com

Abstract

Background

Parasite persistence after acute infection with *Trypanosoma cruzi* is an important factor in the development of Chagas disease (CD) cardiomyopathy. Few studies have investigated the clinical effectiveness of CD treatment through the evaluation of cardiological events by long term follow-up of treated children. Cardiological evaluation in children is challenging since features that would be diagnosed as abnormal in an adult's ECG may be normal, age-related findings in a pediatric ECG trace. The objective was to evaluate cardiac involvement in patients with Chagas disease with a minimum follow-up of 6 years post-treatment.

Methodology

A descriptive study of a cohort of pediatric patients with CD treated with benznidazole (Bz) or nifurtimox (Nf) was conducted. Children (N = 234) with at least 6 years post CD treatment followed at the Parasitology and Chagas Service, Buenos Aires Children's Hospital (Argentina) were enrolled. By convenience sampling, children who attended a clinical visit between August 2015 and November 2019 were also invited to participate for additional cardiovascular studies like 24-hour Holter monitoring and speckle-tracking 2D echocardiogram (STE). Benznidazole was prescribed in 171 patients and nifurtimox in 63 patients. Baseline parasitemia data was available for 168/234 patients.

During the follow-up period, alterations in routine ECG were observed in 11/234 (4.7%, 95% CI [2–7.4%]) patients. In only four patients, with complete right bundle branch block (cRBBB) and left anterior fascicular block (LAFB), ECG alterations were considered probably related to CD.



G OPEN ACCESS

Citation: González NL, Moscatelli G, Moroni S, Ballering G, Jurado L, Falk N, et al. (2022) Longterm cardiology outcomes in children after early treatment for Chagas disease, an observational study. PLoS Negl Trop Dis 16(12): e0010968. https://doi.org/10.1371/journal.pntd.0010968

Editor: Andrea Angheben, Sacro Cuore Hospital, ITALY

Received: July 21, 2022

Accepted: November 21, 2022

Published: December 19, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010968

Copyright: © 2022 González et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data will be available upon approval by the ethics committee of Hospital de Niños Ricardo Gutierrez (email address comite. bioetica.guti@gmail.com) to investigators who meet the criteria for access to confidential data.

Funding: This work was supported by "Fundación para el Estudio de las Infecciones Parasitarias y Enfermedad de Chagas" (FIPEC) Foundation and Drugs for Neglected Diseases initiative (DNDi). DNDi received financial support for this work from UK aid, UK and Fundación Mundo Sano, Argentina; and for its overall mission from Médecins sans Frontières (MSF) and the Swiss Agency for Development and Cooperation (SDC), Switzerland. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

During follow-up, 129/130 (99%) treated patients achieved persistent negative parasitemia by qPCR. Also decrease in *T.cruzi* antibodies titers was observed in all patients and negative seroconversion occurred in 123/234 (52%) patients.

Conclusions

A low incidence of cardiological lesions related to CD was observed in patients treated early for pediatric CD. This suggests a protective effect of parasiticidal treatment on the development of cardiological lesions and highlights the importance of early treatment of infected children.

Trial registration

ClinicalTrials.gov NCT04090489.

Author summary

If left untreated, CD evolves into a chronic oligosymptomatic infection that can progress to cardiac complications in 30% of patients after several years.

It is known that the main early marker of cardiac involvement are alterations in the conduction system. There are few studies of long-term after treatment cardiological evolution which have assessed the clinical effectiveness of treatment.

The rationale for CD treatment is to avoid the development of cardiological complications. The parasiticidal effect of treatment has been demonstrated but its clinical effectiveness in preventing cardiac involvement requires long term follow-up.

In our long term follow-up study of treated children, we observed the preventive effect on cardiac lesions by treatment with benznidazole or nifurtimox. This intensifies the need for early diagnosis and treatment to prevent the development of long term complications observed in CD.

Introduction

Chagas disease (CD), caused by *Trypanosoma cruzi* (*T. cruzi*), has a broad range of hosts, and is primarily acquired in childhood through contact with infected triatominae bugs or congenitally. [1] Historically considered a regional disease, CD has now become a global phenomenon due to migration, reaching regions such as Europe, the USA and Japan. [2–5]

CD presents with an acute phase with high parasitemia that, if left untreated, evolves into a chronic oligosymptomatic infection that can progress, in 30% of patients [6], to cardiac and/or gastrointestinal complications after several years. Parasite persistence is important to the development of CD cardiomyopathy. Effective antiparasitic treatment of CD could, therefore, prevent CD cardiac complications. However, there is insufficient data to adequately evaluate the impact of early CD treatment on the prevention of cardiac lesions, since this requires long-term follow-up of treated pediatric patients.

Electrocardiogram (ECG) alterations are early signs of CD cardiac involvement. Some ECG hallmarks of CD are cRBBB, QRS complex widening or fragmentation, and sinus bradycardia. [6,7] New tests like speckle-tracking 2D echocardiogram (STE) has been developed for assess

myocardial contractility deformation. STE is used to detect subtler segmental ventricular wall motion abnormalities an early marker of cardiovascular compromise in CD patients. [8,9]

Few studies have assessed the cardiological outcomes after early treatment of CD in childhood, possibly because cardiological evaluation in children must be conducted by pediatric cardiologist specialists to avoid misdiagnosing age-related findings in pediatric ECG traces as abnormal. Awareness of these differences is the key to correct interpretation of pediatric ECGs.

The aim of this study is to evaluate the incidence of cardiological changes in CD children several years after treatment completion, by ECG, ambulatory 24-hour-ECG monitoring (Holter) as well as the speckle-tracking 2D echocardiogram (STE) to evaluate systolic and diastolic ventricular contractility. Also kinetics of *T. cruzi* antibodies and parasitemia were evaluated as a marker of treatment response.

Methods

Ethics statement

The study was approved by the ethics committee of the Hospital de Niños Ricardo Gutiérrez, Buenos Aires, Argentina (CEI N°16.29). Written informed consent by the patient and/or parents was obtained for all patients according to age and local regulations. The protocol was registered in ClinicalTrials.gov, NCT04090489.

The identity of the patients was kept confidential, as well as the results of the studies carried out by means of a code of numbers and letters.

A descriptive study of cardiological, parasitological and serological results in a cohort of pediatric CD patients, with at least 6 years' post-treatment follow-up, treated with benznidazole (Bz) or nifurtimox (Nf) was conducted at the Servicio de Parasitología y Chagas, Hospital de Niños Ricardo Gutiérrez, Buenos Aires, Argentina. Additionally, by convenience sampling, children who attended a clinical visit between August 2015 and November 2019 were also invited to participate for additional cardiovascular studies (Holter and STE) (Fig 1).

Also, functional gastrointestinal disorders were evaluated using the Rome IV questionnaire. This questionnaire includes alarm symptoms or red flags to alert clinicians to consider additional testing to investigate the presence of megaesophagus or megacolon.

Since the standard of care for CD is the treatment of all infected children, it was not possible to include a non-treated control group due to ethical issues.

Diagnosis criteria for CD: In infants younger than 8 months, CD was diagnosed by direct observation of T. cruzi using a parasitological concentration method (microhematocrit test—MH) or xenodiagnosis, and in infants older than 9 months, by two reactive serological tests (ELISA, and indirect hemagglutination—IHA).

Inclusion criteria

- Children with CD treated with Bz or Nf with at least 6 years of post-treatment follow-up.
- Signed informed consent or assent according to patient age.

Exclusion criteria

- Patients with chronic diseases (renal, hepatic, neurological) that could affect the interpretation of the results (at the discretion of the researcher).
- Subjects with congenital heart disease.





https://doi.org/10.1371/journal.pntd.0010968.g001

Treatment

Bz (5–8 mg/kg in 2 or 3 doses) or Nf (10–12 mg/kg in 2 or 3 doses) was prescribed for 60–90 days at the time of diagnosis. Medication was provided in monthly batches and compliance was assessed by counting the remaining tablets at each visit.

T. cruzi serology and qPCR

Blood samples were collected at diagnosis, end-of-treatment and every 6–12 months thereafter. Detection of *T. cruzi*-satellite DNA region by PCR was carried out in blood (2 mL) mixed with EDTA-guanidine buffer (10). Serological tests were performed by indirect hemagglutination (IHA) (Chagatest HAI; Wiener lab, Rosario, Argentina) and ELISA (Chagatest ELISA lisado; Wiener lab, Rosario, Argentina tests).

Cardiological evaluation

ECG was performed at diagnosis and every year following treatment, as part of post-treatment follow up protocols. Additionally, by convenience sampling, children who attended a clinical visit between August 2015 and November 2019 were also invited to participate for additional cardiovascular studies (24 hours ECG-Holter and STE).

Evaluation of myocardial strain by STE was performed with a Siemens Acuson SC 2000 echograph and a 2–5MHz cardiac and three-dimensional matrix transducer. The analysis measured cardiac dimensions and cardiac ventricular function using two-dimensional M-mode, Doppler and tissue Doppler imaging with emphasis on measuring longitudinal strain in the left ventricle by speckle tracking. Two-dimensional apical four-chamber view was used to measure global left ventricular peak systolic longitudinal strain and longitudinal systolic strain rate. At least four beats were recorded for later measurements. Encirclement of the ventricle was manually done at the endocardial level. Three-dimensional left ventricle ejection fraction and volume were automatically calculated using Siemens software, in four-chamber views.

Cardiological studies were carried out at the Cardiology service of the Hospital de Niños Ricardo Gutiérrez, by trained pediatric cardiology specialists and pediatric electrophysiologists who were blind to the patients' treatment history.

Statistical analysis

Continuous variables are presented as means with CI_{95%} or medians and interquartile range, and categorical variables as percentages.

The kinetics of *T. cruzi* serum antibodies were analyzed using survival analysis. Analyses were performed with R software v3.0 (R Core Team 2018, R Foundation for Statistical Computing, Vienna, Austria https://www.R-project.org/).

Results

A total of 234 patients (110 males, 124 females) with at least 6 years follow-up were evaluated. The median age at diagnosis was 62.53 months (range 1–202 months, IQR 11–116), with a median follow up time of 9.8 years (6–20.2 years, IQR 7.5–12.4) (Fig 1).

During the follow-up period, alterations in routine ECG were observed in 11/234 (4.7%, 95% CI [2–7.4%]).

However, in 4/11 patients these ECG variations were considered normal variations for children. In the remaining 7/11 patients the ECG variations were considered pathological and only 4/7 patients with cRBBB and LAFB were considered to have ECG abnormalities probably related to CD. Table 1.

In addition, by convenience sampling, a total of 95/234 (40.59%) patients who attended their annual clinical check-up between August 2015 and November 2019 were enrolled for a more in-depth cardiological evaluation with Holter, Doppler echocardiography and STE. 7/11 patients with alterations in routine ECG are found in this group.

Holter monitoring showed pathological findings in 4/95 (4.2%; 95% CI [1.3–11%]) patients. The pathological findings were: isolated ventricular extrasystoles and nocturnal sinus bradycardia (one patient); asymptomatic and vagal related 1st and 2nd degree AV block (one patient); LAFB (one patient); and cRBBB (one patient). Only the last two findings were considered probably related to CD involvement. In these 4 patients, in order to stress myocardial function, ergometer exercise tests with oxygen consumption estimation were conducted, showing normal results.

Another 3/95 (3.1%) patients presented non-pathological findings such as iRBBB).

Myocardial contractility was evaluated by STE in 79/95 (83%) patients (16 patients were not available for evaluation due to missed appointments). STE results: Mean four-chamber global longitudinal strain was -19% while strain rate gave -1 (seg-1). The median longitudinal strain was 19.19%, while for strain rate was -1 (seg-1). Values were found to be above (i.e., less negative) -16% strain in 8 patients. As regards strain rate, 3 patients presented values above

| Age at treatment | Age at evaluation | Treatment | Serology at final evaluation | ECG findings | Other cardiological evaluation | Interpretation |
|---------------------|-------------------|-------------------------------|--|--|--------------------------------|---------------------------------------|
| 3 years | 19 years | Bz 70 days Good compliance | ELISA / IHA: Negative | Complete right bundle branch block | STE: Normal | Pathological, probably related to CD. |
| 11 years | 27 years | Bz 60 days Good compliance | ELISA: Reactive 3.1 IHA: 32 | Complete right bundle branch block | - | Pathological, probably related to CD. |
| 11 years | 22 years | Bz 60 days Good compliance | ELISA: Reactive 3.3 IHA: Negative PCR: Negative | Isolated ventricular extrasystoles and nocturnal sinus bradycardia | STE: Normal | Pathological, non related to CD. |
| 5 years | 12 years | Nf 60 days Good compliance | ELISA / IHA: Negative PCR: Negative | 2nd degree AV block | Echocardiogram: Normal | Pathological, non related to CD. |
| 1 years | 11 years | Bz 60 days Good compliance | ELISA / IHA: Negative PCR: Negative | Asymptomatic and vagal related 1st and 2nd degree AV block | STE: Normal | Pathological, non related to CD. |
| 2 years | 13 years | Bz 60 days Good compliance | ELISA / IHA: Negative PCR: Negative | Incomplete right bundle branch block | STE: Normal | Normal pediatric ECG variations. |
| 1 years | 7 years | Bz 60 days Good compliance | ELISA / IHA: Negative PCR: Negative | Incomplete right bundle branch block | STE: Normal | Normal pediatric ECG variations. |
| 10 months | 10 years | Bz 60 days Good compliance | ELISA / IHA: Negative PCR: Negative | Incomplete right bundle branch block | STE: Normal | Normal pediatric ECG variations. |
| 7 years | 13 years | Bz 60 days Good compliance | ELISA: Reactive: 5.9 / IHA: 16 PCR: Negative | Incomplete right bundle branch block | Echocardiogram: Normal | Normal pediatric ECG variations. |
| 6 years | 16 years | Bz 60 days Good compliance | ELISA: Reactive: 1.8 / IHA: Negative PCR: Negative | Left anterior fascicular block | STE: Normal | Pathological, probably related to CD. |
| 10 years | 16 years | Bz 80 days Good compliance | ELISA: Reactive: 11.9 / IHA: 512 PCR: Negative | Left anterior fascicular block | Echocardiogram: Normal | Pathological, probably related to CD. |

| Table 1. | Description | of patients with | ECG alterations. |
|----------|-------------|------------------|------------------|
|----------|-------------|------------------|------------------|

https://doi.org/10.1371/journal.pntd.0010968.t001

-0.8 (seg-1). These cases were considered normal since the global echocardiographic contractility evaluation was normal.

Left ventricular ejection fraction was measured by three-dimensional echocardiography. Mean ejection fraction was 59.2% (45 to 74%).

In conclusion, no disturbances in myocardial contractility were observed by STE and threedimensional echocardiography. Furthermore, no clinical symptoms such as ventricular tachycardia, syncope, chest pain or palpitations were observed during the follow-up.

Evaluation of parasiticidal treatment response

In the 234 patients, Bz was prescribed in 171 patients and Nf in 63 patients. Mean Bz dose was 6.6 mg/kg per day (range 4.6–9 mg/kg/day) divided in two (n = 156) or three doses (n = 15). Mean Nf dose was 11.28 mg/kg per day (range 7.2–15 mg/kg/day) divided in two (n = 48) or three doses (n = 15). Only 3 patients did not receive a complete treatment for ChD.

Baseline parasitemia data were available for 168/234 patients: 140/168 (83.3%) were initially positive (91 by qPCR, 7 by both qPCR and MH, 31 by MH and 11 by xenodiagnosis), depending on the availability of parasitological tests. 130/140 positive patients were followed up by parasitemia. In 129/130 (98.5%) parasitemia became negative by T. cruzi—qPCR at the end of treatment and remained persistently negative throughout the follow-up.



Fig 2. Serological seroconversion by ELISA and IHA profile in 234 treated patients (Kaplan-Meier).

https://doi.org/10.1371/journal.pntd.0010968.g002

A decrease in *T. cruzi* antibody titers by conventional serology (IHA and ELISA) was observed after treatment. Both serological tests became negative in 123 patients (52.56%) at a median time of 8,59 years (95% CI: 6.97–11.86 years). Fig 2.

Regarding functional gastrointestinal disorders, Rome IV diagnostic questionnaire did not show any relevant gastrointestinal symptoms that would require additional gastrointestinal evaluation in any patients.

Discussion

There are few studies that have investigated the clinical effectiveness of CD treatment, by means of the evaluation of cardiological events, in a long-term follow-up of asymptomatic treated children.

To our knowledge this is the first study of a cohort of treated patients where parasitological, serological and cardiological results, using the latest cardiological available techniques, were evaluated in conjunction with one another in a long term follow-up. A decrease in *T. cruzi* antibodies measured by conventional serology and persistent non-detection of *T. cruzi* by qPCR during after-treatment follow-up was observed in CD treated patients, with an average post-treatment follow-up of 10 years. These results showed the parasiticidal effect of treatment and are in line with previous studies published by our group [10,11], and also studies of other groups [12,13].

No related CD changes in cardiac function, evaluated by ECG and 24-hour ambulatory ECG monitoring and STE, were found in the majority of treated patients. It is likely that this was related to the protective effect of treatment at an early stage of *T. cruzi* infection.

Currently, CD treatment with Bz or Nf is recommended as the standard of care for acute and early chronic phase in children. In the early phase of infection, which is more easily observed in children, the majority of cases are asymptomatic and without cardiac involvement, as described in this study.

Multiple mechanisms may be responsible for the development of cardiac lesions in *T. cruzi* infection. The presence and persistence of the parasite is a critical factor which triggers a specific immune response, inducing vascular endothelial cell damage. [14] The treatment of infected subjects is, therefore, based on the elimination of intracellular parasites to avoid the development of future cardiological complications. [15–17] Several studies have shown that in

non-treated adult CD patients [18–23], the rate of cardiac disease progression was between 13 and 25% with a 1.48 to 4% annual progression rate.

There are some controversial results about the clinical effect of treatment in adults. The most relevant question about CD treatment is whether it will have a clinical benefit preventing the development of cardiological lesions. A pivotal study by Viotti [18] described the beneficial treatment effect in adult patients with chronic CD, changing the perspective about the clinical effect of CD treatment. Others studies, mainly retrospective, showed positive effects of treatment in the progression of heart disease. [13,19,20,24–26]

Recently a retrospective cohort study reinforced previous findings showing that treated patients had fewer ECG alterations (7.9% vs 21.1%) compared with the non-treated group. [27].

However, in the BENEFIT study [28] no difference between treated subjects and the placebo group was observed in adult patients with advanced chronic CD related cardiac complications. This study showed that in advanced chronic CD the cardiological damage was probably irreversible and not modifiable by the parasiticidal treatment.

Regarding children, the available information is based on outdated individual studies and reports vary substantially due to diverse design, small sample size and quality of the studies. However, some positive results about treatment benefits have been published. [12,13,29]

As regards the evaluation of cardiac compromise by ECG it is essential to take into account that children are different from adults. Conduction intervals (PR interval, QRS duration) are shorter than adults due to smaller cardiac size. First degree AV block and Wenckebach sequence may be a normal finding in healthy children during rest or under vagal tone. The QT interval depends on heart rate and age. Heart rates are much faster in neonates and infants, decreasing as the child grows older. Thus, cardiological evaluation in children should be conducted by pediatric cardiologists. In addition, some pathological findings could be related to technical issues and in other cases may be related to congenital lesions and not produced by *T. cruzi* infection.

Awareness of these differences is the key factor for the correct interpretation of pediatric ECG. [30] For example, iRBBB can be a normal ECG alteration during infancy and early childhood. [31] A prevalence of 0.75% for cardiac conduction disturbance including 0.32% prevalence of iRBBB and 0.11% cRBBB was reported in children without any clinical significance. [32] A controversial long-term follow-up study of treated children with Bz for 60 days was not associated with fewer ECG abnormalities as compared to no treatment. Also, the prevalence of ECGs with abnormalities was higher among treated children compared with those not treated [33]. However, it should be emphasized that most of the ECG alterations described were normal findings in children (i.e. extrasystoles, iRBBB, first-degree atrioventricular block) which are unlikely to be related to CD. Only 4/94 children showed alterations of right bundle conduction that could be attributed to CD. This is a clear example of inadequate evaluation of ECG in children, as was seen in several published studies reporting ECG abnormalities in children, which are unlikely to be related to CD. [29,34,35]

Recently a retrospective evaluation of treated children showed differences in ECG alterations between treated and non-treated (4/41 (9.75%) vs 9/41 (21.95%)) patients [36]. However, in contrast to our study, only very few patients were evaluated by 2D echocardiogram or Holter.

In our cohort, with a longer after-treatment follow-up, only four patients showed ECG alterations that could possibly be related to *T. cruzi* infection (cRBBB, LAFB). Regarding the patients with LAFB, it can have many possible causes, and it is not as pathognomonic alterations of CHD. There are other causes that could justify the appearance of these ECG alterations. However, the functional echocardiographic studies were normal. The other ECG

findings observed were in fact likely to be normal ECG findings in children or therefore unrelated to CD. In order to confirm these results, the patients will continue in a prospective follow-up.

New techniques, like STE, have been developed to measure the regional strain percentage of deformation between two ventricle sites during contraction, and strain rate is the speed at which this deformation occurs. In our study using strain and strain rate measurement techniques, treated patients did not show significant long-term regional wall motion. To our knowledge, there were no previous reports of these measurements by STE in pediatric CD treated patients.

Functional gastrointestinal disorders were described in approximately 10% of patients after several years of infection. In order to evaluate gastrointestinal dysfunction, the Rome IV diagnostic questionnaire was used. No relevant gastrointestinal symptoms that required further gastrointestinal evaluation in any patient were found.

A weakness of our study is that it lacks an untreated control group due to the fact that the standard of care for pediatric CD is the treatment of infected children as soon as possible after diagnosis. Also not all treated patients were available for by Holter and STE evaluation. However, our study shows the best available evidence that cardiomyopathy progression rate was lower than in previously reported studies [23,27], reinforcing the preventive treatment effect of CD.

As was previously reported by our group [10,37,38], treated patients showed parasitological and serological signs of treatment response (a decrease in antibody titers, seroreversion in more than half of the patients and constant negative T. cruzi—qPCR). More importantly, we have shown, using the most sensitive cardiological techniques available, that the development of cardiac alterations was most probably prevented by parasiticidal treatment after a median follow-up of over a decade. These results further underline the importance of early diagnosis and availability of treatment, especially for children, young adults and women of childbearing age, in order to prevent new infections and the development of symptomatic cardiac chronic CD.

Further studies with long-term follow-up (20–30 years) of the present cohort will be conducted in order to confirm that early treatment prevents later cardiological compromise in CD patients.

Supporting information

S1 File. STROBE Statement. Checklist of items that should be included in reports of cohort studies. (DOC)

Author Contributions

Conceptualization: Nicolás Leonel González, Jaime Altcheh.

Formal analysis: Nicolás Falk, Facundo Garcia Bournissen.

Investigation: Nicolás Leonel González, Griselda Ballering, Laura Jurado.

Methodology: Nicolás Leonel González.

Resources: Guillermo Moscatelli, Samanta Moroni, Andrés Bochoeyer, Alejandro Goldsman, María Grippo.

Supervision: Nicolás Leonel González, Héctor Freilij, Jaime Altcheh.

Visualization: Nicolás Leonel González, Eric Chatelain, Jaime Altcheh.

Writing - original draft: Nicolás Leonel González, Jaime Altcheh.

Writing – review & editing: Nicolás Leonel González, Guillermo Moscatelli, Samanta Moroni, Griselda Ballering, Laura Jurado, Nicolás Falk, Andrés Bochoeyer, Alejandro Goldsman, María Grippo, Héctor Freilij, Facundo Garcia Bournissen, Eric Chatelain, Jaime Altcheh.

References

- 1. Altcheh JM, Freilij H. Chagas Disease: A Clinical Approach. Springer Nature; 2019. 356 p.
- Abras A, Ballart C, Fernández-Arévalo A, Pinazo M-J, Gascón J, Muñoz C, et al. Worldwide Control and Management of Chagas Disease in a New Era of Globalization: a Close Look at Congenital Trypanosoma cruzi Infection. Clinical Microbiology Reviews [Internet]. 2022 Apr 20; 35(2):e0015221. Available from: https://pubmed.ncbi.nlm.nih.gov/35239422/
- Basile L, Jansa JM, Carlier Y, Salamanca DD, Angheben A, Bartoloni A, et al. Chagas disease in European countries: the challenge of a surveillance system. Euro Surveill [Internet]. 2011 Sep 15; 16(37). Available from: https://www.ncbi.nlm.nih.gov/pubmed/21944556
- Lidani KCF, Andrade FA, Bavia L, Damasceno FS, Beltrame MH, Messias-Reason IJ, et al. Chagas Disease: From Discovery to a Worldwide Health Problem. Frontiers in Public Health [Internet]. 2019 Jul 2; 7(166). Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6614205/
- 5. Bern C, Messenger LA, Whitman JD, Maguire JH. Chagas Disease in the United States: a Public Health Approach. Clinical Microbiology Reviews. 2019 Nov 27; 33(1).
- Prata A. Clinical and epidemiological aspects of Chagas disease. Lancet Infect Dis. 2001 Sep; 1(2):92– 100.
- Rojas LZ, Glisic M, Pletsch-Borba L, Echeverría LE, Bramer WM, Bano A, et al. Electrocardiographic abnormalities in Chagas disease in the general population: A systematic review and meta-analysis. PLoS Negl Trop Dis. 2018 Jun; 12(6):e0006567.
- Acquatella H, Asch FM, Barbosa MM, Barros M, Bern C, Cavalcante JL, et al. Recommendations for Multimodality Cardiac Imaging in Patients with Chagas Disease: A Report from the American Society of Echocardiography in Collaboration With the InterAmerican Association of Echocardiography (ECO-SIAC) and the Cardiovascular Imaging Department of the Brazilian Society of Cardiology (DIC-SBC). J Am Soc Echocardiogr. 2018 Jan; 31(1):3–25.
- Nunes MCP, Badano LP, Marin-Neto JA, Edvardsen T, Fernández-Golfín C, Bucciarelli-Ducci C, et al. Multimodality imaging evaluation of Chagas disease: an expert consensus of Brazilian Cardiovascular Imaging Department (DIC) and the European Association of Cardiovascular Imaging (EACVI). Eur Heart J Cardiovasc Imaging. 2018 Apr 1; 19(4):459–60n.
- Moscatelli G, Moroni S, García Bournissen F, González N, Ballering G, Schijman A, et al. Longitudinal follow up of serological response in children treated for Chagas disease. PLoS Negl Trop Dis. 2019 Aug; 13(8):e0007668.
- Altcheh J, Castro L, Dib JC, Grossmann U, Huang E, Moscatelli G, et al. Prospective, historically controlled study to evaluate the efficacy and safety of a new paediatric formulation of nifurtimox in children aged 0 to 17 years with Chagas disease one year after treatment (CHICO). PLoS Negl Trop Dis. 2021 Jan; 15(1):e0008912.
- Estani SS, Sosa Estani S, Porcel BM, Segura EL, Yampotis C, Ruiz AM, et al. Efficacy of chemotherapy with benznidazole in children in the indeterminate phase of Chagas' disease [Internet]. Vol. 59, The American Journal of Tropical Medicine and Hygiene. 1998. p. 526–9. Available from: <u>https://doi.org/10.4269/aitmh.1998.59.526</u>
- de Andrade AL, Zicker F, de Oliveira RM, Almeida Silva S, Luquetti A, Travassos LR, et al. Randomised trial of efficacy of benznidazole in treatment of early Trypanosoma cruzi infection. Lancet. 1996 Nov 23; 348(9039):1407–13.
- Dávila DF, Rossell O, de Bellabarba GA. Pathogenesis of chronic chagas heart disease: parasite persistence and autoimmune responses versus cardiac remodelling and neurohormonal activation [Internet]. Vol. 32, International Journal for Parasitology. 2002. p. 107–9. Available from: https://doi.org/10. 1016/s0020-7519(01)00311-3
- 15. Zhang L, Tarleton RL. Parasite persistence correlates with disease severity and localization in chronic Chagas' disease. J Infect Dis. 1999 Aug; 180(2):480–6.

- Duffy T, Bisio M, Altcheh J, Burgos JM, Diez M, Levin MJ, et al. Accurate real-time PCR strategy for monitoring bloodstream parasitic loads in chagas disease patients. PLoS Negl Trop Dis. 2009 Apr 21; 3 (4):e419.
- Rassi A Jr, Rassi A, Little WC, Xavier SS, Rassi SG, Rassi AG, et al. Development and validation of a risk score for predicting death in Chagas' heart disease. N Engl J Med. 2006 Aug 24; 355(8):799–808.
- Viotti R, Vigliano C, Armenti H, Segura E. Treatment of chronic Chagas' disease with benznidazole: clinical and serologic evolution of patients with long-term follow-up. Am Heart J. 1994 Jan; 127(1):151– 62.
- 19. Fabbro DL, Streiger ML, Arias ED, Bizai ML, del Barco M, Amicone NA. Trypanocide treatment among adults with chronic Chagas disease living in Santa Fe city (Argentina), over a mean follow-up of 21 years: parasitological, serological and clinical evolution. Revista da Sociedade Brasileira de Medicina Tropical [Internet]. 2007 Feb 1 [cited 2022 Oct 20]; 40:1–10. Available from: https://www.scielo.br/j/rsbmt/a/VVcJ6mCNb9CTGyW9bvxnkCS/?lang=en
- Fragata-Filho AA, França FF, Fragata C da S, Lourenço AM, Faccini CC, Costa CA de J. Evaluation of Parasiticide Treatment with Benznidazol in the Electrocardiographic, Clinical, and Serological Evolution of Chagas Disease. PLoS Negl Trop Dis. 2016 Mar; 10(3):e0004508.
- 21. Pinto Dias JC. Natural history of Chagas disease. Arq Bras Cardiol. 1995 Oct; 65(4):359–66.
- Maguire JH, Hoff R, Sherlock I, Guimarães AC, Sleigh AC, Ramos NB, et al. Cardiac morbidity and mortality due to Chagas' disease: prospective electrocardiographic study of a Brazilian community. Circulation. 1987 Jun; 75(6):1140–5.
- Sabino EC, Ribeiro AL, Salemi VMC, Di Lorenzo Oliveira C, Antunes AP, Menezes MM, et al. Ten-Year Incidence of Chagas Cardiomyopathy Among Asymptomatic Trypanosoma cruzi–Seropositive Former Blood Donors [Internet]. Vol. 127, Circulation. 2013. p. 1105–15. Available from: http://dx.doi.org/10. 1161/circulationaha.112.123612
- Viotti R, Vigliano C, Lococo B, Bertocchi G, Petti M, Alvarez MG, et al. Long-term cardiac outcomes of treating chronic Chagas disease with benznidazole versus no treatment: a nonrandomized trial. Ann Intern Med. 2006 May 16; 144(10):724–34.
- 25. Machado-de-Assis GF, Diniz GA, Montoya RA, Dias JCP, Coura JR, Machado-Coelho GLL, et al. A serological, parasitological and clinical evaluation of untreated Chagas disease patients and those treated with benznidazole before and thirteen years after intervention. Mem Inst Oswaldo Cruz. 2013 Nov; 108(7):873–80.
- Cardoso CS, Ribeiro ALP, Oliveira CDL, Oliveira LC, Ferreira AM, Bierrenbach AL, et al. Beneficial effects of benznidazole in Chagas disease: NIH SaMi-Trop cohort study. PLoS Negl Trop Dis. 2018 Nov; 12(11):e0006814.
- Hasslocher-Moreno AM, Saraiva RM, Sangenis LHC, Xavier SS, de Sousa AS, Costa AR, et al. Benznidazole decreases the risk of chronic Chagas disease progression and cardiovascular events: A longterm follow up study. EClinicalMedicine. 2021 Jan; 31:100694.
- Morillo CA, Marin-Neto JA, Avezum A, Sosa-Estani S, Rassi A, Rosas F, et al. Randomized Trial of Benznidazole for Chronic Chagas' Cardiomyopathy [Internet]. Vol. 373, New England Journal of Medicine. 2015. p. 1295–306. Available from: http://dx.doi.org/10.1056/nejmoa1507574
- de Andrade AL, Zicker F, Rassi A, Rassi AG, Oliveira RM, Silva SA, et al. Early electrocardiographic abnormalities in Trypanosoma cruzi-seropositive children. Am J Trop Med Hyg. 1998 Oct; 59(4):530–4.
- Goodacre S. ABC of clinical electrocardiography: Paediatric electrocardiography [Internet]. Vol. 324, BMJ. 2002. p. 1382–5. Available from: http://dx.doi.org/10.1136/bmj.324.7350.1382
- Chan TC, Sharieff GQ, Brady WJ. Electrocardiographic Manifestations: Pediatric ECG [Internet]. Vol. 35, The Journal of Emergency Medicine. 2008. p. 421–30. Available from: http://dx.doi.org/10.1016/j. jemermed.2007.09.039
- Chiu SN, Wang JK, Wu MH, Chang CW, Chen CA, Lin MT, et al. Cardiac Conduction Disturbance Detected in a Pediatric Population [Internet]. Vol. 152, The Journal of Pediatrics. 2008. p. 85–9. Available from: http://dx.doi.org/10.1016/j.jpeds.2007.05.044
- Colantonio LD, Prado N, Segura EL, Sosa-Estani S. Electrocardiographic Abnormalities and Treatment with Benznidazole among Children with Chronic Infection by Trypanosoma cruzi: A Retrospective Cohort Study. PLoS Negl Trop Dis. 2016 May; 10(5):e0004651.
- 34. Bianchi F, Cucunubá Z, Guhl F, González NL, Freilij H, Nicholls RS, et al. Follow-up of an asymptomatic Chagas disease population of children after treatment with nifurtimox (Lampit) in a sylvatic endemic transmission area of Colombia. PLoS Negl Trop Dis. 2015 Feb; 9(2):e0003465.
- 35. Rangel-Flores H, Sánchez B, Mendoza-Duarte J, Barnabé C, Brenière FS, Ramos C, et al. Serologic and parasitologic demonstration of Trypanosoma cruzi infections in an urban area of central Mexico: correlation with electrocardiographic alterations. Am J Trop Med Hyg. 2001 Dec; 65(6):887–95.

- 36. Suasnábar S, Olivera LV, Arias E, Bizai ML, Bottasso O, Arias E, et al. Trypanocidal therapy among children infected by Trypanosoma cruzi. Serological and electrocardiographic changes over a mean twenty-five-years follow-up period. Acta Trop. 2021 Oct; 222:106050.
- Moscatelli G, Moroni S, García-Bournissen F, Ballering G, Bisio M, Freilij H, et al. Prevention of congenital Chagas through treatment of girls and women of childbearing age. Mem Inst Oswaldo Cruz. 2015 Jun; 110(4):507–9.
- Berenstein AJ, Falk N, Moscatelli G, Moroni S, González N, Garcia-Bournissen F, et al. Adverse Events Associated with Nifurtimox Treatment for Chagas Disease in Children and Adults. Antimicrob Agents Chemother [Internet]. 2021 Jan 20; 65(2). Available from: http://dx.doi.org/10.1128/AAC.01135-20